Author's response to reviews

Title: Spatial distribution of the risk of dengue fever in southeast Brazil, 2006-2007

Authors:

  Ricardo Cordeiro (ricacordeiro@gmail.com)
  Maria R Donalisio (rita.donalisio@gmail.com)
  Valmir R Andrade (valmir@sucen.sp.gov.br)
  Ana CN Mafra (anacarl.nunes@gmail.com)
  Luciana B Nucci (lbnucci@terra.com.br)
  John C Brown (jcbrown2@ku.edu)
  Celso Stephan (celso@stephan.mus.br)

Version: 5 Date: 16 May 2011

Author's response to reviews:

This is the final revised manuscript.